Author's response to reviews

Title: Clinical characteristics and quality of life in women with COPD: an observational study

Authors:

Chantal Raherison (chantal.raherison@isped.u-bordeaux2.fr)
Isabelle Tillie-Leblond (itillieleblond@hotmail.fr)
Anne Prudhomme (ablprudhomme@qmail.com)
Camille Taille (camille.taille@bch.aphp.fr)
Elisabeth Biron (e.biron@free.fr)
Cecilia Nocent-Ejnaini (cnocent@ch-cotebasque.fr)
Benigne Mathieu (benigne.mathieu@astrazeneca.com)
Juliette Ostinelli (Juliette.Ostinelli@astrazeneca.com)

Version: 3 Date: 13 December 2013

Author's response to reviews:

Bordeaux, the 13th december 2013
Dear Editor,

Thanks very much regarding your editorial work. We found the remarks of the reviewers very helpful.

You’ll find enclosed a point-by-point response and a new version of the manuscript revised.

We hope that this second version of the manuscript is now suitable for publication in the BMC Women Health Journal.

Best regards

Chantal Raherison, MD, PhD
Reviewer: Takashi Hajiro

Minor Essential Revisions.

There are some spelling mistakes, typos, and grammatical errors. Please revise the manuscript carefully. Some of them are described below:

Page 2, line 7: pulmonologist # pulmonologists
R1: We replace pulmonologist by pulmonologists

Page 2, line 10: HADS can be eliminated, since the authors did not mention it later in the Abstract.
R2: the term HADS had been eliminated
Page 2, line 11: A space before Results is not necessary.
R3: space before Results has been deleted

Page 4, line 2: became # become
R4: “become” replace “became”

Page 4, line 12: (5)(6) # (5,6)
R5: the correction has been done

Page 4, line 18: HRQoL should be spelled out.
R6: the correction has been done

Quality of written English: Needs some language corrections before being Published
R7: The manuscript has been reviewed by a native English speaker (Ray Cooke).